Author’s response to reviews

Title: Different therapeutic approaches on quality of life in patients with inflammatory bowel disease

Authors:

Junjie Xu (walter235@126.com)
Hui Lin (hui1in1000@126.com)
Xu Feng (xuefeng3@126.com)
Minyue Tang (minyuetang@126.com)
Jun Shen (shenjun_med@yeah.net)
Zhihua Ran (ranzhihua62@sina.cn)

Version: 3
Date: 15 October 2014

Author’s response to reviews: see over
Dear Editor,

On behalf of my co-authors, I am submitting this revised manuscript “Different therapeutic approaches on quality of life in patients with inflammatory bowel disease” for consideration and possible publication as an original research in BMC Gastroenterology. This study aims to validate the mainland Chinese translation of the Inflammatory Bowel Disease Questionnaire (MCIBDQ), and to evaluate the impact of infliximab treatment on HRQOL in patients with inflammatory bowel disease (IBD) for the first time in China, as compared with other therapies of different levels. Furthermore, the impact of different medical therapies on marriage, employment and economic burden in IBD patients were also evaluated.

After we carefully revised this manuscript, here, we submit the point-by-point description of the changes we’ve made as follows:

1. Please include the description of the additional files in the main manuscript.
Response: We’ve added the additional files in the main manuscript and marked them in red.

Best Regards.

Yours Sincerely

Jun Shen and Zhi Hua Ran
State Key Laboratory for Oncogenes and Related Genes, Key Laboratory of Gastroenterology & Hepatology, Ministry of Health, Division of Gastroenterology and Hepatology, Ren Ji Hospital, School of Medicine, Shanghai Jiao Tong University, Shanghai Cancer Institute, Shanghai Institute of Digestive Disease, 145 Middle Shandong Road, Shanghai 200001, China.